Author's response to reviews

Title: Five-year trends in epidemiology and prevention of mother-to-child HIV transmission, St. Petersburg, Russia: results from perinatal HIV surveillance

Authors:

Dmitry M Kissin (DKissin@cdc.gov)
Michele G Mandel (MMandel@cdc.gov)
Natalia Akatova (Akatova_Natasha@hotmail.com)
Nikolay A Belyakov (Akatova_Natasha@hotmail.com)
Aza G Rakhmanova (Rakhmanova@peterlink.ru)
Evgeny E Voronin (EVoronin1@rambler.ru)
Galina V Volkova (Akatova_Natasha@hotmail.com)
Alexey A Yakovlev (Akatova_Natasha@hotmail.com)
Denise J Jamieson (DJamieson@cdc.gov)
Charles Vitek (CVitek@cdc.gov)
Joanna Robinson (Joanna@pedaids.org)
William C Miller (Bill_Miller@unc.edu)
Susan Hillis (SHillis@cdc.gov)

Version: 6 Date: 21 October 2011

Author's response to reviews:

Major revisions were made: Figure numbers were removed from all figures.

Some minor revisions were made: Vertical lines were removed from the table.

We appreciate if you could crop the figures to remove white space around them, if any. Thank you!